\begin{thebibliography}{10}

\bibitem{04Aitken}
CGG Aitken and D~Lucy.
\newblock {Evaluation of trace evidence in the form of multivariate data}.
\newblock {\em {JOURNAL OF THE ROYAL STATISTICAL SOCIETY SERIES C-APPLIED
  STATISTICS}}, {53}({Part 1}):{109--122}, {2004}.

\bibitem{06Aitken}
CGG Aitken, D~Lucy, G~Zadora, and JM~Curran.
\newblock {Evaluation of transfer evidence for three-level multivariate data
  with the use of graphical models}.
\newblock {\em {COMPUTATIONAL STATISTICS \& DATA ANALYSIS}},
  {50}({10}):{2571--2588}, {JUN 20} {2006}.

\bibitem{Bezdek2000}
JC~Bezdek and LI~Kuncheva.
\newblock {Some notes on twenty one (21) nearest prototype classifiers}.
\newblock In {Ferri, FJ and Inesta, JM and Amin, A and Pudil, P}, editor, {\em
  {ADVANCES IN PATTERN RECOGNITION}}, volume {1876} of {\em {LECTURE NOTES IN
  COMPUTER SCIENCE}}, pages {1--16}. {2000}.

\bibitem{Bezdek2001}
JC~Bezdek and LI~Kuncheva.
\newblock {Nearest prototype classifier designs: An experimental study}.
\newblock {\em {INTERNATIONAL JOURNAL OF INTELLIGENT SYSTEMS}},
  {16}({12}):{1445--1473}, {DEC} {2001}.

\bibitem{pca}
C.M Bishop.
\newblock {\em Pattern Recognition and Machine Learning}.
\newblock New York, NY, Springer, 2006.

\bibitem{Cano2005}
José~Ramón Cano, Francisco Herrera, and Manuel Lozano.
\newblock Stratification for scaling up evolutionary prototype selection.
\newblock {\em Pattern Recognition Letters}, 26(7):953 -- 963, 2005.

\bibitem{Chang74}
Chin-Liang Chang.
\newblock Finding prototypes for nearest neighbor classifiers.
\newblock {\em Computers, IEEE Transactions on}, C-23(11):1179--1184, Nov.
  1974.

\bibitem{Hart67}
T.~Cover and P.~Hart.
\newblock Nearest neighbor pattern classification.
\newblock {\em Information Theory, IEEE Transactions on}, 13(1):21--27, Jan
  1967.

\bibitem{Dasarathy94}
B.V. Dasarathy.
\newblock Minimal consistent set (mcs) identification for optimal nearest
  neighbor decision systems design.
\newblock {\em Systems, Man and Cybernetics, IEEE Transactions on},
  24(3):511--517, Mar 1994.

\bibitem{Devi2002}
V.~Susheela Devi and M.~Narasimha Murty.
\newblock An incremental prototype set building technique.
\newblock {\em Pattern Recognition}, 35(2):505 -- 513, 2002.

\bibitem{knn}
Richard~O. Duda and Peter E.Hart.
\newblock {\em Pattern classification and scene analysis}.
\newblock A Wiley-Interscience Publication, New York: Wiley, 1973.

\bibitem{Evett84}
Ian Evett.
\newblock A quantitative theory for interpreting transfer evidence in criminal
  cases.
\newblock {\em Applied Statistics}, 33(1):25--32, 1984.

\bibitem{Evett90}
Ian Evett and John Buckleton.
\newblock The interpretation of glass evidence. a practical approach.
\newblock {\em Journal of the Forensic Science Society}, 30(4):215--223, 1990.

\bibitem{Evett94}
Ian Evett and J.~Lambert.
\newblock Further observations on glass evidence interpretation.
\newblock {\em Science and Justice}, 35(4):283--289, 1995.

\bibitem{fastmap}
Christos Faloutsos and King-Ip Lin.
\newblock Fastmap: a fast algorithm for indexing, data-mining and visualization
  of traditional and multimedia datasets.
\newblock In {\em SIGMOD '95: Proceedings of the 1995 ACM SIGMOD international
  conference on Management of data}, pages 163--174, New York, NY, USA, 1995.
  ACM.

\bibitem{Frank+Asuncion:2010}
A.~Frank and A.~Asuncion.
\newblock {UCI} machine learning repository, 2010.

\bibitem{Garain2008}
Utpal Garain.
\newblock Prototype reduction using an artificial immune model.
\newblock {\em Pattern Anal. Appl.}, 11(3-4):353--363, 2008.

\bibitem{García2008}
Salvador García, José~Ramón Cano, and Francisco Herrera.
\newblock A memetic algorithm for evolutionary prototype selection: A scaling
  up approach.
\newblock {\em Pattern Recognition}, 41(8):2693 -- 2709, 2008.

\bibitem{Grove80}
D.M. Grove.
\newblock Interpretation of forensic evidence using a likelihood ratio.
\newblock {\em Biometrika}, 67(1):243--246, April 1980.

\bibitem{Hart68a}
P.~Hart.
\newblock The condensed nearest neighbor rule (corresp.).
\newblock {\em Information Theory, IEEE Transactions on}, 14(3):515 -- 516, may
  1968.

\bibitem{jalali08}
O.~Jalali, T.~Menzies, and M.~Feather.
\newblock Optimizing requirements decisions with keys.
\newblock In {\em Proceedings of the PROMISE 2008 Workshop (ICSE)}, 2008.
\newblock Available from \url{http://menzies.us/pdf/08keys.pdf}.

\bibitem{joll02}
I.~Jolliffe.
\newblock {\em Principal component analysis}.
\newblock Springer-Verlag, 175 Fifth Avenue, NY, USA, 2002.

\bibitem{Karslake09}
N.~Karslake, S.~Lewis, and W.~Bronswijk.
\newblock Characterisation of automotive paint clear coats by atr-fr-ir with
  subsequent chemometric analysis.
\newblock 2009.

\bibitem{Kim2003}
SW~Kim and BJ~Oommen.
\newblock {A brief taxonomy and ranking of creative prototype reduction
  schemes}.
\newblock {\em {PATTERN ANALYSIS AND APPLICATIONS}}, {6}({3}):{232--244}, {DEC}
  {2003}.

\bibitem{Kohonen90}
T.~Kohonen.
\newblock Improved versions of learning vector quantization.
\newblock pages 545 --550 vol.1, jun 1990.

\bibitem{Kohonen98}
Teuvo Kohonen and Panu Somervuo.
\newblock Self-organizing maps of symbol strings.
\newblock {\em Neurocomputing}, 21(1-3):19 -- 30, 1998.

\bibitem{02Koons}
RD~Koons and J~Buscaglia.
\newblock {Interpretation of glass composition measurements: the effects of
  match criteria on discrimination capability}.
\newblock {\em {JOURNAL OF FORENSIC SCIENCES}}, {47}({3}):{505--512}, {MAY}
  {2002}.

\bibitem{99Koons}
RD~Koons and JA~Buscaglia.
\newblock {The forensic significance of glass composition and refractive index
  measurements}.
\newblock {\em {JOURNAL OF FORENSIC SCIENCES}}, {44}({3}):{496--503}, {MAY}
  {1999}.

\bibitem{01cod}
F~Korn, BU~Pagel, and C~Faloutsos.
\newblock {On the ``dimensionality curse{''} and the ``self-similarity
  blessing{''}}.
\newblock {\em {IEEE TRANSACTIONS ON KNOWLEDGE AND DATA ENGINEERING}},
  {13}({1}):{96--111}, {JAN-FEB} {2001}.

\bibitem{Li2009}
Y~Li, M~Xie, and T~Goh.
\newblock {A study of project selection and feature weighting for analogy based
  software cost estimation}.
\newblock {\em Journal of Systems and Software}, 82:241--252, 2009.

\bibitem{77Lindley}
DV~LINDLEY.
\newblock {PROBLEM IN FORENSIC-SCIENCE}.
\newblock {\em {BIOMETRIKA}}, {64}({2}):{207--213}, {1977}.

\bibitem{09NAS}
Committee on~Identifying the Needs of the Forensic Sciences
  Community;Committee~on Applied and National Research~Council.
  Theoretical~Statistics.
\newblock {\em Strengthening Forensic Science in the United States: A Path
  Forward}.
\newblock National Academies Press, 500 Fifth Street, N.W., Lockbox 285,
  Washington, DC 20055, 2009.

\bibitem{Seheult78}
Allan Seheult.
\newblock On a problem in forensic science.
\newblock {\em Biometrika}, 65(3):646--648, December 1978.

\bibitem{Veenman2005}
CJ~Veenman and MJT Reinders.
\newblock {The nearest subclass classifier: A compromise between the nearest
  mean and nearest neighbor classifier}.
\newblock {\em {IEEE TRANSACTIONS ON PATTERN ANALYSIS AND MACHINE
  INTELLIGENCE}}, {27}({9}):{1417--1429}, {SEP} {2005}.

\bibitem{Walsh94}
K.~Walsh, J.~Buckleton, and C.~Triggs.
\newblock A practical example of the interpretation of glass evidence.
\newblock {\em Science and Justice}, 36(4):213--218, 1996.

\bibitem{09aZadora}
G.~Zadora.
\newblock {Evaluation of evidence value of glass fragments by likelihood ratio
  and Bayesian Network approaches}.
\newblock {\em {ANALYTICA CHIMICA ACTA}}, {642}({1-2, Sp. Iss. SI}):{279--290},
  {MAY 29} {2009}.

\bibitem{09Zadora}
G.~Zadora and T.~Neocleous.
\newblock {Likelihood ratio model for classification of forensic evidence}.
\newblock {\em {ANALYTICA CHIMICA ACTA}}, {642}({1-2, Sp. Iss. SI}):{266--278},
  {MAY 29} {2009}.

\end{thebibliography}

%\section{Introduction}
Since the creation of the Nearest Neighbour algorithm in 1967 (Hart \cite{Hart67}), a copious amount of prototype learning schemes (PLS) have appeared to remedy the five (5) major drawbacks associated with the algorithm and it's variations. First, the high computation costs caused by the need for each test sample to find the distance between it and each training sample. Second, the storage requirement is large since the entire dataset needs to be stored in memory. Third, outliers can negatively affect the accuracy of the classifer. Fourth the negative effect of data sets with non-separable and/or overlapping classes and last, the low tolerance to noise. To solve these issues, PLS are used. Their main purpose is to reduce a training set via various selection and/or creation methods to produce \emph{good prototypes}. \emph{Good} here meaning that the prototypes are a good representation of the original training data set such that they maintain comparable or improved performance of a nearest neighbour classifier while simultaneously eliminating the effects of the drawbacks mentioned previously.

A review of the literature on prototype learning for this thesis has yielded at least 40 PLS each indicating with experimental proof that their particular design is comparable or better than the standard schemes; published surveys of PLS can be found in \cite{Bezdek2000, Bezdek2001, Kim2003}. However many of these schemes suffer from at least one of the following disadvantages:

\bi
\item computationally expensive
\item order effects (where the resulting prototypes are unreasonably affected by the order of the original training set)
\item overfitting
\ei

The goal of this thesis is not to be unduly critical of the PLS which may succumb to any of the above disadvantages (particularly since many of them have proven to be successful), but rather to present a novel approach to this field of study which overcomes these disadvantages to report little or no loss in recognition of NNC.

Thus, this thesis presents CLIFF, a prototype learning scheme which runs in linear time, is independent of the order of the training set and avoids overfitting. A novel feature of CLIFF is that instead of either removing or adding (un)qualified prototypes from/to a 'prototype list', a conjunction of contraints is generated for each target class. These conjuctions are created using a ranking algoritm call BORE (Best Or Rest) \cite{jalali08}, which basically finds a range of values for each attribute which best represents the specific target class i.e. have the highest ranks. Any of the instances in the training set which adheres to all the constraints is or are selected as prototypes. Using this structure the percentage reduction of the training set is related to the number of constraints used on the prototype selection process, in that the more constraints used the lower the percentage reduction. So there is no need to predetermine the number of prototypes desired when using CLIFF.

After describing the design and operation of CLIFF, its performance is demonstrated by evaluating it using cross-validation experiments with the wide variety of standard data sets from the UCI repository \cite{Frank+Asuncion:2010}. Our experiments also use FastMap \cite{fastmap} \footnote{A Fast Algorithm for Indexing, Data Mining, and Visualization of Traditional and Multimedia Datasets} and Feature Subset Selection (FSS) to avoid the curse of dimensionality \cite{01cod} which negatively affects Nearest Neighbor Classifiers (NNC). 

Next we describe how CLIFF can be used as part of a tool/model for the evaluation of trace forensic evidence. The principal goal of forensic evaluation models is to check that evidence found at a crime scene is (dis)similar to evidence found on a suspect. In our studies of forensic models for evaluation particularly in the sub-field of glass forensics, we conjecture that many of these models succumb to the following flaws:

\be
\item A tiny error(s) in the collection of data;
\item Inappropriate statistical assumptions, such as assuming that the distributions of refractive indices of glass collected at a crime scene or a suspect obeys the properties of a normal distribution; 
\item and the use of measured parameters from surveys to calculate the \emph{frequency of occurrence} of trace evidence in a population
\ee

In this work we show that CLIFF plays an effective role in the evaluation of forensic trace evidence.

Our research is guided by the following research question:

\begin{itemize}
\item Is CLIFF viable as a Prototype Learner for NNC? \newline
The goal here is to see if the performance of CLIFF is comparable or better than the plain k nearest neighbor (KNN) algorithm. So in our first experiment we compare the performance of predicting the target class using the entire training set to using only the prototypes generated by CLIFF.

%\item Can CLIFF be successful as part of a forensic evaluation model? \newline
%With this forensic case study we introduce the concept of \emph{brittleness} to forensic models. \emph{Brittleness} here is described as tiny changes in input data could lead to a massive change in the output. So our goal here is to show that by incorporating CLIFF as part of a forensic model that \emph{brittleness} will be reduced thereby improving the results of the model.
\end{itemize}

\section{Statement of Thesis}

\section{Contributions of this Thesis}

\section{Structure of this Thesis}

The remaining chapters of this thesis are structured follows:

\bi
\item Chapter 2 provides a survey of Prototype Learning Schemes over the years
\item Chapter 3 describes the design and operation of CLIFF
\item Chapter 4 describes the data sets used along with the preprocessing tools used to avoided the curse of dimensionality \cite{01cod}
\item Chapter 5 presents a detailed description of the experimental procedure followed to analyze the data using CLIFF
\item Chapter 6 examines a case study in which CLIFF is used to reduce the \emph{brittleness} of a forensic model
\item Chapter 7 conclusions are presented
\ei
